Title: Case Report of Ocular Kaposi's Sarcoma

Authors:
Jing Yang (hiyangjing101246@hotmail.com)
Xiaofang Yin (jiegen521@163.com)
Yongping Li (yongpingli1961@aliyun.com.cn)
Shiyou Zhou (zhoushiy@mail.sysu.edu.cn)

Version: 1 Date: 14 May 2017

Author’s response to reviews:

BOPH-D-16-00328

Case Report of Ocular Kaposi's Sarcoma

Jing Yang; Xiaofang Yin; Yongping Li; Shiyou Zhou

BMC Ophthalmology

Dear Editor and reviewers,

Based on your suggestion, we revised our manuscript "Case Report of Ocular Kaposi's Sarcoma" (BOPH-D-16-00328) point by point.

We made the following changes to our manuscript, listed below:

1. Background

Response: we have added the background section to the manuscript as you suggested.

2. Please also make the following revisions to your Declarations section:

   1). Please change the text under your Ethics section to state, "not applicable".

   2). Please revise your Competing interests section to clarify if any financial or non-financial affiliations influenced this study.
3). Please revise your Authors' contribution section to confirm all authors have read and approved the final version of your manuscript.

4). Please add a Funding section to the Declarations, and please state any funding received for your study. If no funding was received, please state so.

5). Please upload all figures as separate figure files and remove them from the main manuscript. Please do not include the figure captions within the figure files, but instead place them at the end of the main manuscript in a section titled “Figure legends”.

Response: we have made changes to the manuscript according to the questions above.